Author’s response to reviews

Title: Effect of a typical systemic hospital reform on inpatient expenditure for rural population: the Sanming model in China

Authors:
Zhaolin Meng (mengzhaolin2005@126.com)
Min Zhu (zmmin2002@126.com)
Yuanyi Cai (caiyuanyi_541@163.com)
Xiaohong Cao (xhcao@cmu.edu.cn)
Huazhang Wu (hzwu@cmu.edu.cn)

Version: 5 Date: 22 Mar 2019

Author’s response to reviews:

22 Mar, 2019
Dear Dr Shona Reeves and Dr Joseph Minus,

Thank you very much for your thoughtful suggestions to our manuscript "Effect of a typical systemic hospital reform on inpatient expenditure for rural population: the Sanming model in China" (BHSR-D-18-01894).

We have revised the article according to the suggestions.

1. We re-phrased the textual overlap.
2. We removed the duplicate additional files list after the Conclusion section.
3. We uploaded our manuscript as a single, final, clean version.

All authors have read and approved the revised manuscript, and there are no financial or other relations that could lead to a conflict of interest.

If you have any further question or any suggestion about our revised manuscript, please contact me. Thank you very much for your input.
We look forward to hearing from you.

Yours sincerely,

Huazhang Wu